Author’s response to reviews

Title: Fetal weight estimation at term – Ultrasound versus clinical examination with Leopold’s manoeuvres: a prospective blinded observational study

Authors:

Oliver Preyer (oliver.preyer@tauernklinikum.at)
Heinrich Husslein (heinrich.husslein@meduniwien.ac.at)
Nicole Concin (nicole.concin@i-med.ac.at)
Anna Ridder (anna.ridder@hotmail.com)
Maciej Musielak (maciej.musielak@tauernklinikum.at)
Christian Pfeifer (christian.pfeifer2@chello.at)
Willi Oberaigner (willi.oberaigner@gmail.com)
Peter Husslein (peter.husslein@meduniwien.ac.at)

Version: 2 Date: 05 Mar 2019

Author’s response to reviews:

Dear Prof. Coore,

thank you very much for your kind suggestions as they helped to improve the quality of our manuscript. Point by point responses to the comments are listed in the following:

1. Our manuscript was additionally language edited by a certified translator (Mrs. Mary Heaney Margreiter) again.

2. We clearly indicated the approval of our institutional review board (IRB) / ethics committee and stated this in the ethics approval and consent to participate section.
3. The consent was verbal and was documented in the patient’s records. We confirmed and stated this in the methods section of our manuscript.

4. We included figure legends with the requested informations in the revised version of our manuscript.

5. We changed figure 2 to a multi-panel figure a-e on a single A4 page.

6. We acknowledge the services of Mary Heaney Margreiter for professional language editing.

7. We provide a list of abbreviations after the conclusions section in the revised version of our manuscript.

8. & 9. n/a